Author’s response to reviews

Title: Encephalitis temporally associated with Japanese encephalitis vaccine: Case report

Authors:

Na Jia (jiana79_41@hotmail.com)
Qiu Min Zhao (prettynana@sohu.com)
Xiao Fang Guo (prettynana@sohu.com)
Jing Xia Cheng (prettynana@sohu.com)
Chao Wu (prettynana@sohu.com)
Shu Qing Zuo (prettynana@sohu.com)
Pei Fang Dai (prettynana@sohu.com)
Jun Ying Zhao (prettynana@sohu.com)
Jiu Song Zhang (zhang_jiusong@yahoo.com.cn)

Version: 2 Date: 18 November 2010

Author’s response to reviews:

Dear Editor,

As the suggestions, we have improved the English using a commercial copyediting service.

In addition, we want to clarify we have collected sera and CSF samples from the patients and gained ethical approval to examine the records. We also received parental consent for publication of the cases described and would provide the copy as necessary.

Best regards,

Jiu-Song Zhang